Author’s response to reviews

Title: Implementation of the uterine fibroids Option Grid patient decision aids across five organizational settings: a randomized stepped-wedge study protocol

Authors:

Peter Scalia (peter.scalia@dartmouth.edu)
Marie-Anne Durand (Marie-Anne.Durand@dartmouth.edu)
Rachel Forcino (rachel.forcino@dartmouth.edu)
Danielle Schubbe (danielle.c.schubbe@dartmouth.edu)
Paul Barr (Paul.J.Barr@dartmouth.edu)
Nancy O’Brien (Nancy.L.O’Brien@dartmouth.edu)
A. O’Malley (James.OMalley@dartmouth.edu)
Tina Foster (Tina.C.Foster@hitchcock.org)
Mary Politi (mpoliti@wustl.edu)
Shannon Laughlin-Tommaso (laughlintommaso.shannon@mayo.edu)
Erika Banks (ebanks@montefiore.org)
Tessa Madden (maddent@wustl.edu)
Raymond Anchan (ranchan@bwh.harvard.edu)
Johanna Aarts (Annemijn.Aarts@radboudumc.nl)
Priscilla Velentgas (priscillavelentgas@gmail.com)
Joyce Balls-Berry (BallsBerry.Joyce@mayo.edu)
Carla Bacon (carla@nuff.org)
Monica Adams-Foster (Monica.Adams-Foster@dartmouth.edu)
Carrie Mulligan (ccmhats@gmail.com)
Sateria Venable (sateria@fibroidfoundation.org)
Author’s response to reviews:

Dear Reviewers,

We thank you for giving us the opportunity to re-submit our protocol manuscript. We have addressed your comment by reducing the manuscript word count down to 5,026 - below the 5500 word count limit imposed by Implementation Science. We thank you for your time in considering our work and look forward to hearing from you regarding potential publication in your journal.

Yours sincerely,

Peter Scalia PhD, Postdoctoral Fellow
The Dartmouth Institute for Health Policy and Clinical Practice
Level 5, Williamson Translational Research Building
One Medical Center Drive
Lebanon NH 03756
E: peter.scalia@dartmouth.edu
T: (603) 653-0832